Author's response to reviews

Title: Home: The place the older adult can not imagine living without

Authors:

Catharina Gillsjö (catharina.gillsjo@his.se)
Donna Schwartz-Barcott (dsb@uri.edu)
Iréne von Post (irene.von.post@telia.com)

Version: 5 Date: 17 February 2011

Author's response to reviews: see over
Cover letter to corrected proof MS 1967917546397023, “Home: The place the older adult can not imagine living without.”

Thank you for publishing this manuscript in BMC Geriatrics. This corrected proof is sent since I have not got an updated proof with a new date from the editor production team and there is a due date today in an earlier e-mail that was sent to me. Sorry if I should have waited but I was not sure how to do.

It was discovered when reading the proof that the highlighted yellow and the bolded text that was made to clarify the changes when resubmitting the revision of the manuscript were still present. This has been changed back to normal.

I also found extra spaces between some letters in the manuscript that has been deleted.

There was also a mark for citation on page 6 in the end of a sentence that was removed.

The word well known was sometimes spelled with a hyphen and sometimes not. The hyphen has been taken out in the text on p.20 and in the figure.

Residential facility on page 23 has been changed to residential facilities.

A paragraph in the text on page 18 had been changed into italics like a quote but that is changed back to normal text.

In the beginning paragraph of the results on page 13-14 there has been changes in the use of colon and semicolon. One colon is added when presenting the subthemes to the first main theme. One colon is removed when presenting the second theme and one semicolon is replaced with a colon when presenting the sub-themes to the second theme.

To make sure that the figure text in the end of the main document don´t fall out, the figure text is included in the main document just below the text where the figure 1 should be inserted.

A t has been added in the last name on the front page of the manuscript for the second author.

One reference in the reference list has been changed from being in press to being published: Gillsjö C, Schwartz-Barcott D: A concept analysis of home and its meaning in the lives of three older adults. Int J Older People Nurs 2011, 6(1):4-12.

Sincerely yours,

Catharina Gillsjö, PhD(c), FNP, RN (Corresponding author: catharina.gillsjo@his.se)

Donna Schwartz-Barcott, PhD, RN, Professor

Iréne von Post, PhD, RN, RNA